Author's response to reviews

Title: Application of the COOP/WONCA charts for aged patients with chronic obstructive pulmonary disease in comparison between Japanese and Chinese populations

Authors:

Midori Yamaguchi (tomoe@med.kurume-u.ac.jp)
Motoyuki Nakao (nakao_motoyuki@med.kurume-u.ac.jp)
Hideto Obata (obata@tip.ne.jp)
Hideki Ikeda (sa802308@yahoo.co.jp)
Tetsuro Kanda (sa802308@yahoo.co.jp)
Qiao Wang (wq1225@yahoo.co.jp)
Yoriko Hara (hara01@med.kurume-u.ac.jp)
Hisamitsu Omori (omorih@gpo.kumamoto-u.ac.jp)
Yoko Ishihara (blackcat@med.kurume-u.ac.jp)

Version: 2 Date: 17 January 2013

Author's response to reviews:

BMC Public Health

Application of the COOP/WONCA charts for aged patients with chronic obstructive pulmonary disease in comparison between Japanese and Chinese populations

Dear Editor,

Thank you for your useful comments and suggestions on the structure of our manuscript.

We have modified the manuscript accordingly, and detailed corrections are listed below point by point:

1) Please supply English translations of all questionnaires developed for use in this study, and upload them as additional files.

We have attached English versions of the COOP/WONCA charts and SF-36v2TM as additional files and added “v2TM” to “SF-36” in the manuscript.

The manuscript has been resubmitted to your journal. We look forward to your positive response.

Sincerely,

Yoko Ishihara Ph.D.